Author's response to reviews

Title: Dissemination of information to General Practitioners: a questionnaire survey

Authors:

Padma Moorjani (padma.moorjani@mrchear.man.ac.uk)
Heather M Fortnum (heather.fortnum@nottingham.ac.uk)

Version: 2 Date: 3 November 2004

Author's response to reviews:

We are grateful to the reviewers for their postive comments. Neither suggested major or minor revisions. We have addressed the discretionary revisions suggested by reviewer 1 by adding a sentence at the beginning of the results part of the abstract and adding a paragraph at the end of the main methods section. We have corrected the spelling of Yoshinaga in the references. We have formatted the submission in accordance with guidelines and completed the completing interests section.